Author’s response to reviews

Title: Neuropsychopathological comorbidities in learning disorders

Authors:

lucia margari (lucia.margari@uniba.it)
maura buttiglione (maurabuttiglione@alice.it)
francesco craig (fcraig.npi@gmail.com)
arcanego cristella (arcangelo.cristella@libero.it)
concetta de giambattista (concettadegiambattista@gmail.com)
emilia matera (emy7906@libero.it)
francesca operto (francescaoperto@libero.it)
marta simone (martasimone_ms@libero.it)

Version: 3
Date: 2 December 2013

Author's response to reviews: see over
Dear Editor,

We really thank the reviewer for the useful comments and suggestions that improved our paper “Neuropsychopathological comorbidities in learning disorders”.

We answered all the queries point by point.

Editor comments

“In view of the recently established DSM V, it would be both useful as well as up to date with the important changes introduced with its use, if the authors could add a small addendum (at the end of the Discussion Section), describing shortly how do they think the new classification scheme would alter their results if they would re-design their study today.”

Thanks to the Editor for this important consideration. We have added a new comment about the classification of DSM-5 at the end of Discussion section.

We will be indebted with the Editorial board of BMC Neurology if this revised paper can now be considered for the publication in the Journal.

Many thanks in advance for your courtesy

Lucia Margari